Accelerator Status Report

Run Schedule: April 23, 2009 to June 15, 2009
May 20, 2009

Injector

· The QE is getting low. A spot move is needed soon (in a day or two).

· Injector energy 66.40 MeV

· Wien angle 66.92. Ops changed the Wien.

· After the GUN 2 reactivation QE is stable.

· Injector has been stable since the GUN2 anode was properly grounded.

Accelerator

· LINPAC Setup 590.20 Me
· Pass setup: 2/5/5
· Optics is stable. 

· RF trip rate is low. Routine maintenance on RF is still needed: recovery and tuning. 

· (May 16)VBV8R09 air hose cracked while delivering beam. The valve closed. The faulty hose was replaced.

· (May 17)There were problems with the SL DP HV cabling, which caused two vacuum valves in the SL to close. EES opened the valves and the cable will be replaced during the summer down.

· (May 19) 1L06 insulating vacuum pump failed and the pump was replaced.
· When taking the FOPT data use only one beam. It does not work well with more than one beam at the same time.

· Optics is stable.

· IPM4T09 is not reliable and was disabled.

· The power supply and crate controlled for CAMAC iocch1 were replaced.

· At the end of the Sunday swing shift there was a power loss in building E2, E3 and E4 due to a breaker trip. We do not know why the breaker tripped.
· Monday (5/11) some box supplies tripped off due to problem with the Warwick substation.

· Yesterday’s (5/12) beam studies went well. 

· It appears that RF needs a lot of tuning. Some cavities trip off and come back SOSing. Some RF cavities were bypassed and recovered. 2L21-1 Klystron was replaced. 
Hall A
E05-015/E08-005    2.429 GeV     2 Pass
· Hall A beam was fine.

· Hall A was changed from 3-pass to 2-pass.

· The pass change went well and the Compton rates are good.

· The Compton detector software was updated. Chime turns off the beam whenever the detector moves. If the hall wants to move the detector Ops needs to turn off the Chime and turn it back on after the movement. The viewer position of the detector was disabled.

Hall B
Eg1-DVCS    5.968 GeV    5 Pass
· Two thirds of Hall B line quads are railed due to the high energy run. The hardware and software limit will be adjusted during the summer down.

· Beam image on SLM 2C20 was not centered (located in the upper right-hand corner on the viewer). Some suspected that beam might be scraping. Ops sent tune beam to the hall and the BPMs showed that beam positions were fine.

· Hall B beam was fine.

Hall C

· Running Hall C beam to the BSY dump.

· The hall is not scheduled to run.
